Author's response to reviews

Title: Contextual and Individual Assessment of Dental pain period prevalence in adolescents: a multilevel approach

Authors:

Marco A Peres (mperes@ccs.ufsc.br)
Karen G Peres (karengp@ccs.ufsc.br)
Antonio C Frias (acfrias@usp.br)
José Leopoldo F Antunes (leopoldo@usp.br)

Version: 5 Date: 12 August 2010

Author's response to reviews:

UNIVERSIDADE FEDERAL DE SANTA CATARINA
CENTRO DE CIÊNCIAS DA SAÚDE
DEPARTAMENTO DE SAÚDE PÚBLICA
CAMPUS UNIVERSITÁRIO – TRINDADE
CEP: 88049-970 – FLORIANÓPOLIS – SANTA CATARINA – BRASIL
Florianópolis, August 11, 2010
Natalie Pafitis MSc
The BioMed Central Editorial Team
BMC Oral Health

Dear Natalie,

On behalf of all authors, I am submitting the final version of the manuscript of Contextual and Individual Assessment of Dental pain period prevalence in adolescents: a multilevel approach.

The text has been revised according to all indications of the editorial board of the BMC Oral Health.

Yours sincerely,

Marco Aurélio Peres